Author’s response to reviews

Title: HIV among immigrants living in high-income countries: a realist review of evidence to guide targeted approaches to behavioural HIV prevention

Authors:

Tadgh McMahon (mcmah@email.cs.nsw.gov.au)
Paul R Ward (paul.ward@flinders.edu.au)

Version: 2 Date: 30 July 2012

Author’s response to reviews:

After initial submission we noticed a slight error in the order in which we had presented the results. This entailed a revision of the main manuscript and a re-numbering of two of the Figures.